




\documentclass[11pt,twoside]{article}
\usepackage{amssymb}
\usepackage[mathscr]{eucal}


\setlength{\textheight}{19.5cm}
\setlength{\textwidth}{13cm}
\itemsep=0pt
\parindent=7mm
\parsep=0pt


\def\ii{\'{\i}}                     % i acentuada

\def\R{\mathbb R}                   % Numeros reales
\def\N{\mathbb N}                   % Numeros naturales
\def\Z{\mathbb Z}                   % Numeros enteros
\def\C{\mathbb C}
\def\S{\mathbb S}
\def\L{\mathbb L}


\newtheorem{theorem}{Theorem}[section]
\newtheorem{corollary}[theorem]{Corollary}
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{proposition}[theorem]{Proposition}
\newtheorem{axiom}{Axiom}
\newtheorem{definition}{Definition}[section]
\newtheorem{remark}[theorem]{Remark}

\usepackage{graphicx}

\begin{document}
\begin{thebibliography}{99}


\bibitem{BeeEhrEas96}  \sc J.K. Beem, P.E. Ehrlich and K.L. Easley,
\rm {\em Global Lorentzian Geometry}, Second ed., Pure and Applied Math. {\bf 202}, Marcel Dekker, 1996.

\bibitem{Carriere} \sc Y. Carri\`ere, {\em Autour de la conjeture de L. Markus sur les vari\'et\'es affines},
\rm Inven. Math. {\bf 95} (1989), 615--628.

\bibitem{CT} \sc A.A. Coley and B.O.J. Tupper, {\em Special conformal Killing vector space-times and symmetry inheritance}, {\rm J. Math. Phys.} {\bf 30} (1989), 2616--2625.


\bibitem{Gre63}  \sc L.W. Green,
{\em Auf Wiedersehensfl\"{a}chen},
\rm Ann. of Math. {\bf 78} (1963), 289--299.

\bibitem{GutPal Rom01}  \sc M. Guti\'errez, F.J. Palomo and A. Romero,
{\em A Berger-Green type inequality for Compact Lorentzian Manifolds},
\rm Trans. Amer. Math. Soc. (2002) (to appear).

\bibitem{GutPal Rom02} \sc M. Guti\'errez, F.J. Palomo and A. Romero, {\em Conjugate points along null geodesics on Lorentzian manifolds with symmetries}, \rm  Proc. X Fall Workshop on Geometry and Physics,
Miraflores de la Sierra (Madrid), Pub. R.S.M.E., (2002).

\bibitem{Har82}  \sc S.G. Harris,
{\em A triangle comparison theorem for Lorentz manifolds},
\rm Indiana Univ. Math. J. {\bf 31} (1982), 289--308.

\bibitem{Har85}  \sc S.G. Harris,
{\em A characterization of Robertson-Walker Spaces by null sectional curvature},
\rm Gen. Relativity Gravitation {\bf 17} (1985), 493--498.

\bibitem{Kar82}  \sc H. Karcher,
{\em Infinitesimale Charakterisierung von Friedmann-Universen},
\rm Arch. Math. {\bf 38} (1982), 58--64.

\bibitem{Kli96}  \sc B. Klinger,
{\em Completude des varietes lorentziennes a courbure constante},
\rm Math. Ann. {\bf 306} (1996), 353--370.

\bibitem{KobNom69} \sc S. Kobayashi and K. Nomizu,
{\em Foundations of Differential Geometry} \rm Vol. {\bf 2}, Wiley Inters. Publ., New York, 1969.

\bibitem{Koc85}  \sc L. Koch-Sen,
{\em Infinitesimal null isotropy and Robertson-Walker metrics},
\rm J. Math. Phys. {\bf 26} (1985), 407--410.

\bibitem{Rom} \sc A. Romero, {\em The introduction of Bochner's technique on Lorentzian manifolds},
\rm Nonlinear Anal. {\bf 47} (2001), 3047--3059.

\bibitem{RomSan94b}  \sc A. Romero and M. S\'{a}nchez,
{\em On completeness of compact Lorentzian manifolds},
\rm Geometry and Topology of Submanifolds {\bf VI}, \rm {World Scientific}, (1994), 171--182.

\bibitem{RomSan94c} \sc A. Romero and M. S\'{a}nchez,
{\em On Completeness of Certain Families of Semi-Riemannian Manifolds},
\rm Geom. Dedicata {\bf 53} (1994), 103--117.

\bibitem{RomSan95}  \sc A. Romero and M. S\'{a}nchez,
{\em Completeness of compact Lorentz manifolds admiting a timelike conformal-Killing vector field}, \rm Proc. Amer. Math. Soc. {\bf 123} (1995), 2831--2833.

\bibitem{RomSan96}   \sc A. Romero and M. S\'{a}nchez,
{\em An integral inequality on compact Lorentz manifolds and its applications},
\rm Bull. London Math. Soc. {\bf 28} (1996), 509--513.

\bibitem{RomSan98}  \sc A. Romero and M. S\'{a}nchez,
{\em Bochner's technique on Lorentz manifolds and infinitesimal conformal symmetries},
\rm Pacific J. Math. {\bf 186} (1998), 141--148.

\bibitem{SachWu77} \sc R. Sachs and H. Wu,
\rm {\em General Relativity for Mathematicians},
Springer-Verlag, 1977.

\bibitem{Yurt90}  \sc U. Yurtsever, {\em Test fields on compact space-times},
\rm  J. Math. Phys. {\bf 31} (1990), 3064--3077.

\end{thebibliography}


\pagestyle{myheadings}
\markboth
{{\sc Conjugate Points along null geodesics}}
{{\sc M. Guti\'{e}rrez, F.J. Palomo and A. Romero  }}
\vspace{1cm}

\thispagestyle{empty}

\begin{center}

{\Large{\bf{A New Approach to the Study of Conjugate Points}}}

\medskip

{\Large {\bf{along Null Geodesics on Certain Compact}}}

\medskip

{\Large{\bf{Lorentzian Manifolds}}}

\vskip1cm

{\sc Manuel Guti\'{e}rrez$^1$, Francisco J. Palomo$^1$ and Alfonso Romero$^2$}
\vskip0.5cm

{\it $^1$ Departamento de \'{A}lgebra, Geometr\'{\i}a y Topolog\'{\i}a,
Universidad de M\'{a}laga}

\vskip0.3cm
{\it $^2$ Departamento de Geometr\'{\i}a y Topolog\'{\i}a,
Universidad de Granada}
\end{center}

\begin{center}
E-mail: {\ttfamily mgl@agt.cie.uma.es, fjpalomo@eresmas.com, \linebreak aromero@ugr.es}
\end{center}
\bigskip


\begin{abstract}
An integral inequality on a compact Lorentzian manifold admitting
a timelike conformal vector field is shown under some assumption
on its conjugate points along null geodesics. The inequality
relates the behaviour of these conjugate points to global
geometrical results. As an application,  several properties of the
null geodesics of a natural Lorentzian metric on each odd
dimensional sphere are obtained.
\end{abstract}


\section{Introduction}

\hyphenation{re-mar-ka-ble}

In \cite{GutPal Rom01}, \cite{GutPal Rom02} the authors have
introduced a new integral inequality on a remarkable family of
compact Lorentzian manifolds. It reproves a classical result of M.
Berger and L.W. Green in Riemannian geometry \cite[Theors. 4.2,
5.3]{Gre63}, and, in a suitable way, extends it to the Lorentzian
setting. The main aim of this note is to show the use of that
integral inequality to the study of conjugate points along null
geodesics on Lorentzian odd dimensional spheres. In fact, each odd
dimensional sphere may be endowed with a natural Lorentzian metric
(section $3$). Our method permits us to study  null conjugate
points without using the Jacobi equation and related techniques.
Moreover, as far as we know, there are not many examples of
compact Lorentzian manifolds where the behaviour of their null
geodesics, null conjugate points and null conjugate loci have been
described.

Compact Lorentzian manifolds have been historically neglected
because of both physical and mathematical reasons. Recall that
they have closed timelike curves, and therefore they are acausal
(in particular, they cannot be isometrically immersed in a
Lorentz-Minkowski space of any dimension) and not physically
admissible. On the other hand, a compact Lorentzian manifold may
be geodesically incomplete (this fact is well known) and the
elliptic model of Lorentzian space form is not compact (contrary
to the Riemannian case). However, it has been recently argued
\cite{Yurt90} that the study of field theory on compact spacetimes
could be interesting for Physics and it could give valuable
information about the underlying manifold, complementary to the
one obtained from the Riemannian theory. From a mathematical point
of view, the lack of completeness in the compact case gave rise to
the obtention of extra conditions which joint to compactness would
imply completeness of the Lorentzian manifold. For instance, in
\cite {Kli96} it has been proved that every compact Lorentzian
manifold with constant sectional curvature is geodesically
complete (the flat case was previously shown in \cite{Carriere});
in \cite{RomSan95} that every compact Lorentzian manifold which
admits a timelike conformal vector field is geodesically complete
(see also \cite{RomSan94b} for a wide information on completeness
of Lorentzian manifolds). Physicists are familiarized with the
study of conformal vector fields, in fact the assumption of their
existence on spacetime is a way to impose some symmetry useful,
for instance, to study the Einstein equations (see, for example
\cite{CT}). Finally, recall the outstanding role of timelike
conformal vector fields in the introduction of Bochner's technique
in Lorentzian manifolds \cite{RomSan96}, \cite{RomSan98},
\cite{Rom}.

The content of this note is organized as follows. Section 2 is first devoted to recall the notion and main properties of the null congruence associated to a timelike conformal vector field on a Lorentzian manifold. In the compact case, an integral inequality is shown, Theorem \ref{Theorem2}, and, using a well known result of H. Karcher, it is analyzed when the equality holds. Moreover, we also show that Theorem \ref{Theorem2} provides information on the manifold from the nonexistence of null conjugate points.

Finally, in section 3 we consider a natural Lorentzian metric $g$ on each sphere $\S^{2n+1}$ (it was called {\it canonical} in \cite{Yurt90}) which is introduced from three different procedures. It is shown that $g$ has a large group of isometries,  an isotropic property for null tangent directions and that it is homogeneous, Proposition \ref{pro4.2a}. Its null geodesics are studied, showing that no null geodesic is closed. Null conjugate points and null conjugate loci are analyzed, Proposition \ref{cor4.3}. In fact, it is shown that all past (or future) null geodesics starting from a point $p$ meet at the second conjugate point of $p$, and that the null conjugate locus at every point is an imbedded $(2n-1)-$dimensional sphere $\S^{2n-1}$.

\hyphenation{na-tu-ral}

\section{Preliminaries}

Let $(M,g)$ be an $n(\geq 2)-$dimensional Lorentzian manifold; that is a
(connected) smooth manifold $M$ endowed with a non-degenerate metric $g$
with index $1$, i.e. with signature $(-,+,...,+)$. As usual, $T_pM$ denotes the tangent space at $p\in M$, $TM$ the tangent bundle of $M$, and $\pi :TM\longrightarrow M$ the natural projection. We shall write $\nabla $ for the Levi-Civita
connection of $g$, $R$ for the Riemannian curvature tensor (our convention on the
curvature tensor is $R(X,Y)Z=\nabla _{X}\nabla _{Y}Z-\nabla _{Y}\nabla
_{X}Z-\nabla _{[X,Y]}Z$), $\mathrm{Ric}$ for the Ricci tensor, $\widetilde{\rm{Ric}}$ for the corresponding quadratic form, $S$ for the scalar curvature and $d\mu _{g}$ for the canonical measure induced from $g$.

The causal character of a tangent vector $v\in T_{p}M$ is {\it timelike}
(resp. {\it null}, {\it spacelike}) if $g(v,v)<0$ (resp. $g(v,v)=0$ and $v\neq 0,$ $
g(v,v)>0$ or $v=0)$. If $v\in T_{p}M$ then, $\gamma _{v}$ will denote the
unique geodesic such that $\gamma _{v}(0)=p$ and $\gamma _{v}^{\prime }(0)=v$. It is well-known that the causal character
of the velocities $\gamma ^{\prime }(t)$, for
any geodesic $\gamma $ of $(M,g)$, does not depend on the parameter $t$. In
particular, a null geodesic $\gamma $ of $(M,g)$ is a geodesic such that $
\gamma ^{\prime }(t)$ is a null vector. A vector field $K\in \mathfrak{X}(M)$ is
said to be timelike if $K_{p}$ is timelike for all $p\in M$. A timelike or null tangent vector $v \in T_{p}M$ is said to be future (resp. past) with respect to $K$ if $g(v,K_{p})<0$ (resp. $g(v,K_{p})>0$). We will write $U=hK$ where $h=[-g(K,K)^{-\frac{1}{2}}]$ and so $g(U,U)=-1$ holds on all $M$.

Let $\widehat{g}$ be the {\it Sasaki metric} on $TM$ induced from the
Lorentzian metric $g$. We point out that it may be introduced in a similar way to
the Riemannian case. But now $\widehat{g}$ is semi-Riemannian with index 2, and the fact that the natural projection $\pi :(TM,\widehat{g})\longrightarrow (M,g)$ is a semi-Riemannian submersion remains true.

Throughout the remainder of this paper, $(M,g)$ will denote a Lorentzian manifold
with dimension $n\geq 3$, time oriented by a timelike vector field $K$. Recall \cite{Har85}, \cite{Koc85} that the \textit{null congruence associated to }$K$ is defined as follows:

\begin{equation}
C_{K}M=\left\{ v\in TM \, : \, g(v,v)=0, \, g(v,K_{\pi (v)})=1\right\}.
\end{equation}

\noindent This subset of $TM$ has the following nice properties \cite{GutPal Rom01}, \cite{GutPal Rom02}:

\vspace{1mm}

\noindent $(i)$ For each null tangent
vector $v,$ there exists a unique $t\in \R$ such that $tv\in C_{K}M$,
and the map $v\longmapsto \left[ v\right] $ is a diffeomorphism from $C_{K}M$
to the manifold $\mathcal{N}=\left\{ \left[ v\right] \in PM : g(v,v)=0 \right\}$
of the null directions of $M$ (here $PM$ denotes the projective fiber bundle associated to $TM$).

\vspace{1mm}

\noindent $(ii)$ It is an orientable imbedded submanifold of $TM$
with dimension $2(n-1)$. Moreover $(C_{K}M,\pi ,M)$ is a fiber bundle with
fibre type $\S ^{n-2}$, and so $C_{K}M$ will be compact if $M$ is assumed to be compact.

\vspace{1mm}

\noindent $(iii)$ The induced metric on $C_{K}M$
from the Sasaki metric of $TM$, which we agree also to represent by $\widehat{g}$, is Lorentzian.
Moreover, the restriction of $\pi $ to $C_{K}M$ is a semi-Riemannian
submersion with spacelike fibers.

\vspace{1mm}

Sectional curvature of a Lorentzian metric can be defined for nondegenerate tangent planes
but it cannot be stated for null planes (i.e. degenerate planes). If $v$ is a null tangent vector and  $\sigma $ a
null plane containing it, the {\it null sectional curvature} with
respect to $v$ of the plane $\sigma $ is defined to be $\mathcal{K}
_{v}(\sigma )=g(R(u,v)v,u)/g(u,u)$, where $\{u,v\}$ is a basis of $\sigma $ \cite{Har82}, \cite{Har85}, \cite[Def. A.6]{BeeEhrEas96}. Note that $\mathcal{K}
_{v}(\sigma )$ does not depend on the choice of the non-zero spacelike vector
$u,$ but it does quadratically on $v$.



From now on let us suppose that a null congruence associated with a timelike vector field $K$ has been fixed. Then, we may choice,
for every null plane $\sigma $, the unique null vector $v\in C_{K}M\cap \sigma $, thus the null sectional
curvature can be thought as a function on null tangent planes. In this note we
always use such convention, and we will call it the $K-$\textit{normalized null sectional curvature}.

\hyphenation{sa-tis-fies}

Until now, no extra hypothesis on the timelike vector field $K$ has been assumed.  Recall that a vector field $X$ is called {\it conformal} (resp. {\it Killing}) if each of its (local) fluxes consists of (local) conformal (resp. isometric) transformations. It is well known that $X$ is conformal if and only if the Lie derivative of $g$ with respect to $X$ satisfies $\mathcal {L}_Xg=\rho g$, where $\rho : M \rightarrow \R$ (Killing when $\rho =0$). If $K$ is assumed to be conformal, then every null geodesic $\gamma _{v}$ of $(M,g)$
with $v\in C_{K}M$, provides us with the null geodesic $\gamma _{v}^{\prime }$
of $(C_{K}M,\widehat{g})$. Furthermore, each null geodesic $\beta $ of $
(M,g) $ may be reparametrized to obtain a null geodesic $\alpha $ which
satisfies $\alpha ^{\prime }(t)\in C_{K}M.$ In fact, consider the real number $a=g(\beta ^{\prime
},K_{\beta })$, which satisfies $a\neq 0$. If we put $\alpha (t)=\beta (
\frac{t}{a})$, then $g(\alpha ^{\prime },K_{\alpha })=1$ holds for all $t$. Null geodesics will be considered to be parametrized by this $K-${\it affine parameter}.


We will next assume that $(M,g)$ is a compact Lorentzian manifold and $K$ a timelike conformal vector field. Recall that in this case $(M,g)$ is geodesically complete \cite{RomSan95}. The following integral inequality is the key tool to relate null conjugate points to global geometric properties:


\begin{theorem}\label{Theorem2}
{\em\cite[Theor. 3.5]{GutPal Rom01}} Let $(M,g)$ be a compact Lorentzian manifold which admits a
timelike conformal vector field $K$. If there exists $a\in (0,+\infty )$ such that every null
geodesic $\gamma _{v}:\left[ 0,a\right] \longrightarrow M$, with $
v\in C_{K}M$, has no conjugate point of $\gamma _{v}(0)$ in $[0,a)$, then
\begin{equation}
\int_{M}h^{n-2}d\mu _{g}\geq \frac{a^{2}}{\pi ^{2}(n-1)(n-2)}\int_{M}\left[ n
\widetilde{\rm{Ric}}(U)+S\right] h^{n}d\mu _{g}.  \label{30}
\end{equation}
Moreover, equality holds if and only if $(M,g)$ has $U-$normalized null
sectional curvature $\frac{\pi ^{2}}{a^{2}h^{2}}$.
\end{theorem}

Observe that if equality holds in (\ref{30}) then the $U-$normalized
null sectional curvature of $(M,g)$ is an everywhere non-zero
point function. On the other hand, it was proven by H. Karcher, \cite{Kar82} the following result:

\vspace{5mm}


\textit{Let} $U$ \textit{be a unit timelike vector field on an }$n(\geq 4)-$
\textit{dimensional Lorentzian manifold $(M,g)$. The $U-$normalized null sectional curvature is an
everywhere non-zero point function if and only if the following conditions hold:}

\vspace{1mm}

\textit{1. The distribution }$U^{\perp }$\textit{\ is integrable.}

\vspace{1mm}

\textit{2. The integral manifolds of }$U^{\perp }$\textit{\ are totally
umbilic and have constant sectional curvature.}

\vspace{1mm}

\textit{3. }$(M,g)$\textit{\ is locally conformal to a flat Lorentzian manifold.}

\vspace{5mm}


Combining Theorem \ref{Theorem2} and Karcher's result we can give a characterization
of the equality in (\ref{30}) in terms of the distribution $U^{\perp}(=K^{\perp})$
and the locally conformal flatness of $(M,g)$.


Moreover, Theorem \ref{Theorem2} provides also information of $(M,g)$ from the nonexistence of null conjugate points. In fact, if it is assumed that every null geodesic does not contain a pair of mutually conjugate points,
then (\ref{30}) is valid for any positive real number $a$. Therefore it must happen
\begin{equation}
\int_{M}\left[ n\widetilde{\rm{Ric}}(U)+S\right] h^{n}d\mu _{g}\leq 0.
\label{Hopf2}
\end{equation}



In the next section we will use the integral inequality (\ref{30}) to get a bound of $a$
for a relevant family of compact Lorentzian manifolds which admit a unit timelike Killing vector field.

\section{Lorentzian Odd Dimensional Spheres}

We consider $\R ^{2n+2}$ identified to $\C ^{n+1}$ as usual: $
(x_{1},...,x_{2n+2})=(z_{1},...,z_{n+1})$, with
$z_{j}=x_{j}+ix_{n+1+j}$. So that, the unit sphere of $\R ^{2n+2}$ is written
\[
\S ^{2n+1}=\Big\{ z=(z_{1},...,z_{n+1})\in \C
^{n+1}: \sum_{j=1}^{n+1}\left| z_{j}\right| ^{2}=1\Big\}.
\]

Let $U\in \mathfrak{X}(\S ^{2n+1})$ be given by $U_{z}=iz$ at any $
z\in \S ^{2n+1}.$ For the canonical Riemannian metric $g_{can}$ of $\S ^{2n+1}$, $U$ is Killing and satisfies $g_{can}(U,U)=1.$ Therefore, $\nabla _{U}U=0,$
where $\nabla $ is the Levi-Civita connection of $g_{can}$; that is, the
integral curves of $U$ are geodesics of $g_{can}$.

Let $\omega$ be the $1-$form metrically equivalent to $U$ with respect to $g_{can}$. A Lorentzian metric on $\S^{2n+1}$ can be defined by
\begin{equation}\label{38}
g \, = \, g_{can}\, - \, 2\, \omega\otimes\omega.
\end{equation}
This construction of $g$ from $g_{can}$ is standard, but the Lorentzian metric $g$ deserves of making stand out among all the Lorentzian metrics of $\S ^{2n+1}.$ In fact, it has previously considered \cite{Yurt90}. It is not difficult to show that the Levi-Civita connection $\widetilde{\nabla}$ of $g$ satisfies:
\begin{equation}
\widetilde{\nabla}_{X}Y=\nabla _{X}Y-2\, \omega(X)
\nabla_{Y}U-2\, \omega(Y)\nabla _{X}U, \label{conexion}
\end{equation}
where $X,Y \in \mathfrak{X}(\S ^{2n+1})$.
Moreover, the vector field $U$ satisfies
$g(U,U)=-1$, it is Killing for $g$ and $\widetilde{\nabla }_{U}U=0$; so that, its integral curves are
unit timelike geodesics of $g$. On the other hand, observe that the inclusion
$\S^{2n+1} \hookrightarrow \S^{2m+1}$, $n<m$, $(z_{ 1},..,z_{n+1})\mapsto(z_{1},..,z_{n+1},0,..,0)$
is a totally geodesic Lorentzian submanifold, when both spheres are endowed with the corresponding Lorentzian metrics (\ref{38}).

Recall now the classical Hopf fibration
$\Pi :(\S ^{2n+1},g_{can})  \rightarrow  (\C P^{n},g_{FS})$, $z  \mapsto  [z]$, where $\C P^{n}$ is the complex projective space endowed with its
Fubini-Study K\"ahler metric $g_{FS}$ of constant holomorphic sectional curvature $4$
\cite[p. 273]{KobNom69}. Recall that $\Pi$ permits to consider $\S ^{2n+1}$ as a principal fiber bundle
over $\C P^{n}$ with structural group $\S ^{1}$. Moreover, $\Pi$ is a Riemannian submersion with totally geodesic fibres.
If the Riemannian metric $g_{can}$ is replaced by the Lorentzian metric $g$, then $\Pi$
becomes a semi-Riemannian submersion from $(\S ^{2n+1},g)$ to $(\C P^n,g_{FS})$ with timelike totally geodesics fibres. Let us remark that $g$ may be considered as a particular case of a Kaluza-Klein metric. In fact, if we put $\mathfrak{ s}^{1}=i\R$ for the Lie algebra of $\S^{1}$ then $i\omega$ is a connection form on $\S^{2n+1}$, and $g \, =\, \Pi^*(g_{FS})\, -\, \omega\otimes \omega$, \, \cite{RomSan94c}.

As a third description of the Lorentzian metric $g$, note that it can be characterized from the properties:
\begin{equation}
\begin{array}{l}
g_{\mid _{\mathcal{V}}}=-g_{can\mid _{\mathcal{V}}}, \quad
g_{\mid _{\mathcal{H}}}=g_{can\mid _{\mathcal{H}}}, \quad
g(\mathcal{V},\mathcal{H})=0, \label{variacion}
\end{array}
\end{equation}
where $\mathcal{V}$ and $\mathcal{H}$ are respectively the vertical and the
horizontal distributions for the canonical connection of the Hopf fibration.

\vspace{1mm}

Now recall that if $U$ is a unit timelike vector field on a Lorentzian manifold $(M,g)$,
and $p\in M$, $(M,g)$ is said to be \textit{spatially isotropic}
with respect to $U$ at $p$ if for every two unit vectors $u_{1},u_{2}\in
U_{p}^{\bot }$ there exists an isometry $\phi :M\longrightarrow M$ such that $
\phi (p)=p,$ $d\phi _{p}(U_{p})=U_{p}$ and $d\phi _{p}(u_{1})=u_{2}$.
$(M,g)$ is said to be \textit{spatially isotropic} with respect to $U$ if it
is \textit{spatially isotropic} with respect to $U$ at every point,
\cite[p. 47]{SachWu77}. The following results are easy to show:

\begin{lemma}
\label{lem4.1}Let $(M,g)$ be a Lorentzian manifold which admits a unit
timelike vector field $U$. Then $(M,g)$ is spatially isotropic with respect to $U$ if and only
if for every $p\in M$ and for every $u,v\in (C_{U}M)_{p}$ there exists an
isometry  $\phi :M\longrightarrow M$ such
that $\phi (p)=p,$ $
d\phi _{p}(U_{p})=U_{p}$ and $d\phi _{p}(u)=v$.
\end{lemma}


\begin{proposition}\label{pro4.2a}
{\em\cite[Prop. 4.2]{GutPal Rom01}} $(\S ^{2n+1},g)$ is spatially isotropic with respect to $U
$ and the unitary group $U(n+1)$ acts transitively by $g-$isometries on $
\S ^{2n+1}$.
\end{proposition}

Now note that in order to analyze the behaviour of the null
geodesics of the Lorentzian odd dimensional spheres, it suffices
to consider the ones starting from the specific point
$p=(1,...,0)\in\S^{2n+1}$. Observe that $v\in
(C_{U}\S^{2n+1})_{p}$ if and only if $v=(-i,v_2 ,...,v_{n+1})$
with $\sum_{j=2}^{n+1} \mid v_{j} \mid^ 2=1$.

\vspace{1mm}

If we agree to represent $\gamma_v(t) =(\Theta_{1}^{v}(t) ,...,\Theta_{n+1}^{v}(t))$, with $\Theta_{k}^{v} :\R\rightarrow\C$, $1\leq k\leq n+1$, then we get:

\begin{displaymath}
\Theta_{1}^{v}(t)=\frac{2-\sqrt{2}}{4}\, e^{(-2-\sqrt{2})it}+\frac{2+\sqrt{2}}{4}\, e^{(-2+\sqrt{2})it}
\end{displaymath}

\begin{displaymath}
\Theta_{j}^{v}(t)=\frac{\sqrt{2}iv_j}{4}\, \Big[e^{(-2-\sqrt 2)it}-e^{(-2+ \sqrt 2)it}\Big], \;  j \geq 2,
\end{displaymath}

The following figures show each kind of components of a lightlike geodesic.

\begin{figure}[h]
\begin{center}
\includegraphics[width=6cm]{geod1.eps}
\includegraphics[width=5.5cm]{geod2.eps}
\end{center}
\end{figure}



From the previous equations the following facts directly follow:

\vspace{1mm}

(1) {\it There is no closed null geodesic in} $(\S^{2n+1},g)$,

(2) {\it For every} $v, u\in (C_{U}\S^{2n+1})_{p}$, $v\not=u$, $\gamma_{v}(t)=\gamma_{u}(t)$ {\it holds if and only if} $t=\frac{k\pi}{\sqrt{2}}$ {\it for some} $k\in \Z$.

\vspace{1mm}

Now we pay attention to curvature properties of $(\S ^{2n+1}, g)$. Its scalar curvature $S$ can be computed to obtain $S=2n(2n+3)$. On the other hand, we get $\widetilde{\mathrm{Ric}}(U)=2n$ and the $U-$normalized null sectional curvature of $(\S^{2n+1},g)$ is a point function if and only if $n=1$, with $\mathcal{K}_{v}(v^{\perp})=8$ for any $v\in C_{U}\S^{3}$, (see \cite{GutPal Rom01} for details). So, it should be pointed out that the first conclusion in Karcher's theorem does not remain true if it
is assumed $\dim M=3$, because of non integrability of the distribution $U^{\perp}$.

We end this note with an application of our integral inequality (\ref{30}) to the study of the behaviour of conjugate points along null geodesics in Lorentzian odd dimensional spheres.
\begin{proposition}\label{cor4.3}
{\em\cite[Prop. 4.4]{GutPal Rom01}} For every null geodesic $\gamma _{v}$ of $(\S ^{2n+1},g)$ with $
v\in C_{U}\S ^{2n+1},$ the points $\gamma _{v}(0)$ and $\gamma _{v}(\frac{\pi }{2\sqrt{
2}})$ are conjugate and there is no conjugate point to $\gamma _{v}(0)$ on $[0,\frac{\pi }{2\sqrt{2}})$.
Moreover the past null conjugate locus of each point $p\in\S^{2n+1}$ is a $(2n-1)-$dimensional imbedded sphere.
\end{proposition}


Observe that previous result may be dualized to analyze the future null conjugate locus.

\begin{remark}
\rm A conjugate point $\gamma(a)$ of $\gamma(0)=p$ along a null geodesic $\gamma$ can be interpreted as an ``almost-meeting point'' of null geodesics starting from $p$. In our case, the first conjugate point along any null geodesic is exactly at the middle of the path to the first ``meeting point'' which is the second null conjugate point. Thus, null geodesics of $\S^{2n+1}$ have an ``Auf wiedersehensfl\"{a}chen'' type property as in Riemannian case, but in contrast to that, in the Lorentzian setting the  first ``meeting point'' is not the first conjugate point.
\end{remark}

{\section*{Acknowledgments}}
\noindent
The first author has been partially supported by MCYT-FEDER Grant \linebreak BFM2001-1825 and CECJA Grant FQM-213, the second author has been partially supported by CECJA Grant FQM-213, and the third author has been partially supported by MCYT-FEDER Grant BFM2001-2871-C04-01.

\bigskip




\end{document}
